Author's response to reviews

Title: Vasorelaxant effects of Angelica dahurica on isolated rat aortic rings

Authors:

Kyungjin Lee (dostudy@naver.com)
Min Sik Shin (ilshim2001@hanmail.net)
Inhye Ham (iham@khu.ac.kr)
Ho-Young Choi (hychoi@khu.ac.kr)

Version: 2
Date: 7 May 2015

Author's response to reviews: see over
7 May 2015

Dear Journal of BMC Complementary & Alternative Medicine:

Our study was submitted to, and approved by the Kyung Hee University Institutional Animal Care and Use Committee. The approval number is KHUASP[SE]-09-042. We described it in the manuscript (Page 7, line 196, Statement of ethics), “This study was submitted to, and approved by the Kyung Hee University Institutional Animal Care and Use Committee (Approval number: KHUASP[SE]-09-042)”.

Thank you very much again.

With Kind Regards,

Ho-Young Choi.

The corresponding author, his address, and his phone number are as follows:

1 Ho-Young Choi
1 K.M.D., Ph. D., Professor
1 Department of Herbology, College of Korean Medicine
1 Kyung Hee University,
1 26 Kyungheedae-ro, Dongdaemun-gu, Seoul, 130-701, Republic of Korea
1 Tel: + 82 2 9619372 Fax: + 82 2 9659372 E-mail: hychoi@khu.ac.kr

All other authors, their e-mail addresses are as follows:

1 Kyungjin Lee : dostudy@naver.com
1 Min Sik Shin : ilshim2001@hanmail.net
1 Inhye-Ham : iham@khu.ac.kr